Author's response to reviews

Title: An uncommon case of random fire-setting behavior associated with Todd paralysis: a case report.

Authors:

Masayuki Kanehisa (kanehisa@oita-u.ac.jp)
Katsuhiko Morinaga (morin@oita-u.ac.jp)
Yoshihiro Maruyama (maruyama@oita-u.ac.jp)
Taiga Ninomiya (ninomiyataiga@oita-u.ac.jp)
Yoshinobu Ishitobi (istbysnb@oita-u.ac.jp)
Yoshihiro Tanaka (tanaka-yoshihiro@oita-u.ac.jp)
Jusen Tsuru (tsuru@oita-u.ac.jp)
Hiroaki Hanada (hanada-h@oita-u.ac.jp)
Jotaro Akiyoshi (akiyoshi@oita-u.ac.jp)
Tomoya Yoshikawa (to_yoshikawa@ffri.co.jp)

Version: 4 Date: 2 March 2012

Author's response to reviews: see over
Dear Dr. Deesha Majithia

Thank you for your e-mail.

We revised manuscript as your suggestion.

MS: 4523864586157599

Case report
An uncommon case of random fire-setting behavior associated with Todd paralysis
Masayuki Kanehisa, Katsuhiko Morinaga, Yoshihiro Maruyama, Taiga Ninomiya, Yoshinobu Ishitobi, Yoshihiro Tanaka, Jusen Tsuru, Hiroaki Hanada, Jotaro Akiyoshi and Tomoya Yoshikawa
BMC Psychiatry (Section: Substance-related disorders, addiction and impulse control)

<1>
- We just want to clarify that you have uploaded the correct version of your manuscript as there are certain elements missing which were previously there. For example please include the following:

1.0 A competing interests section
2.0 An authors contribution section

We changed the competing interest “Non-financial competing interests” to “Authors have no competing interests to declare that are relevant to the content of this submission”

We change Authors’ contributions “MK and KM participated in the interview and diagnosed. YM and TN participated in EEG study. YI and YT participated in SPECT study, JT and HH participated in psychological test examination, JA conceived of the study, and participated in its design and coordination and helped to draft the manuscript. All authors read and approved the final manuscript” to “MK and KM followed up the patient during the admission. YM and TN participated in EEG study. YI and YT examined SPECT study, HK, JT and HH examined psychological test examination, JA planed the study, and participated in its design and coordination and helped to draft the manuscript. All authors read and approved the final manuscript”

<2>
- Please could you include a consent statement and also clarify whether you had obtained consent to publish the images. A statement to this effect must be included in the manuscript. The manuscript should also include a statement to this effect in the 'Acknowledgements'
section, in follows: "Written consent was obtained from the patient or their relative for publication of study".

We added the consent statement in the 'Acknowledgements' section “Written consent was obtained from their relative for publication of study”

<3>
- We also note that on the tile page you refer to this as short communication when in actual fact it is a case report. Please could you change this to reflect that this is a case report.

We changed the title “An uncommon case of random fire-setting behavior associated with Todd paralysis” to “An uncommon case of random fire-setting behavior associated with Todd paralysis: a case report”

<4>
We changed the style of Abbreviations “the Fourth Edition of the Diagnostic and Statistical Manual of Mental Disorders – Text Revision (DSM-IV-TR); electroencephalogram (EEG); Z-score imaging system (eZIS); magnetic resonance imaging (MRI); The Wechsler Adult Intelligence Scale-III (WAIS-III); Single-photon emission computed tomography (SPECT)” to “”.

<5>
We added the coauthors “Hisae Kohno”

Sincerely yours,

Jotaro Akiyoshi M.D., PhD.
Associate Professor of Psychiatry
Department of Neuropsychiatry
Oita University Faculty of Medicine
Yufu-Shi
Oita 879-5593
Japan

E-mail: akiyoshi@oita-u.ac.jp

TEL: 81-97-586-5823